Author's response to reviews

Title: Benzofuroxan derivatives N-Br and N-I induce intrinsic apoptosis in melanoma cells by regulating AKT/BIM signaling and display anti metastatic activity in vivo.

Authors:

Camyla F Farias (camyla.ff@gmail.com)
Mariana H Massaoka (mari massa@gmail.com)
Natalia Girola (nataliagirola@hotmail.com)
Ricardo A Azevedo (ricardoazevedo@hotmail.com)
Salomao D Jorge (sdjorge@gmail.com)
Leoberto C Tavares (leoberto@usp.br)
Carlos R Figueiredo (rogernty@hotmail.com)
Luiz R Travassos (luiztravassos@gmail.com)
Adilson K Ferreira (ferreira-kleber@usp.br)

Version: 5 Date: 14 October 2015

Author's response to reviews:

In the latest version of the manuscript we have added a section for Abbreviations and corrected the text so that definitions of abbreviations were always added at the first mention. No other alterations were made in the present version.